Author’s response to reviews

Title: Mitogen-stimulated cell proliferation and cytokine production in major depressive disorder patients

Authors:

Ping Lin (linpingsun2000@aliyun.com)
Bingyu Ding (531036503@qq.com)
Yunqiang Wu (872049658@qq.com)
Ke Dong (sjtumicro@gmail.com)
Qingtian Li (qingtianli@sjtu.edu.cn)

Version: 1 Date: 15 May 2018

Author’s response to reviews:

Dear Editor,

Thank you for your directions.

1, We have added the "Declarations" heading.

2, We added the animal based ethical review statement.

3, We added the animal care-taking, the source of animals and the method of sacrifice.

4, We completed and attached the ARRIVE checklist.

Yours QT